Author's response to reviews

Title: Implementation strategies of internet-based asthma self-management support in usual care. Study protocol of the IMPASSE study - a cluster randomized trial

Authors:

Johanna L van Gaalen (j.l.van_gaalen@lumc.nl)
Moira J Bakker (m.j.bakker@lumc.nl)
Letti van Bodegom-Vos (i.vanbodgem-vos@lumc.nl)
Jiska B Snoeck-Stroband (j.b.snoeck-stroband@lumc.nl)
Willem JJ Assendelft (w.j.j.assendelft@lumc.nl)
Ad A Kaptein (a.a.kaptein@lumc.nl)
Christian Taube (c.taube@lumc.nl)
Jacob K Sont (j.k.sont@lumc.nl)

Version: 2 Date: 12 October 2012

Author’s response to reviews:

In the revised manuscript you we have added the Consort checklist for cluster trials. Additionally we have included the requested information on strategy allocation in the manuscript.